Author's response to reviews

Title: Intraindividual changes of dipeptidyl peptidase-IV in peripheral blood of patients with rheumatoid arthritis are associated with the disease activity

Authors:

Lucie Sromova (lsrom@lf1.cuni.cz)
Petr Busek (busekpetr@seznam.cz)
Liliana Sedova (sedo@revma.cz)
Aleksi Sedo (aleksi@cesnet.cz)

Version: 5 Date: 29 August 2015

Author's response to reviews: see over
Cover Letter- response to the editors
MS: 1013314847160766 - Intraindividual changes of dipeptidyl peptidase-IV in peripheral blood of patients with rheumatoid arthritis are associated with the disease activity

Dear Dr. Mockridge and Dr. Singh

We have implemented the name of the institutional ethics committee that approved the study in the methods section and marked them with yellow background.

Sincerely yours,

Prof. Aleksi Šedo, MD, PhD